Author’s response to reviews

Title: A randomized controlled longitudinal naturalistic trial testing the effects of automatic self transcending meditation on heart rate variability in late life depression: study protocol

Authors:

Zareen AMTUL (zamtul@uwo.ca)
Amanda Arena (Amanda.Arena@lhsc.on.ca)
Hussein Hirjee (hhirjee2016@meds.uwo.ca)
Zaineb U. Khan (afaan98@hotmail.com)
Pramudith M. Maldeniya (PramudithManujaya.Maldeniya@londonhospitals.ca)
Ronnie I. Newman (rnewman@artofliving.org)
Amer M. Burhan (Amer.Burhan@sjhc.london.on.ca)
Stephen Wetmore (Stephen.Wetmore@schulich.uwo.ca)
Akshya Vasudev (Akshya.Vasudev@lhsc.on.ca)

Version: 2 Date: 6 August 2014

Author’s response to reviews:

We accept the transfer to BMC Complementary and Alternative Medicine from BMC Psychiatry.